Author's response to reviews

Title: The social validity of a national assessment centre for selection into general practice training.

Authors:

Annette Burgess (annette.burgess@sydney.edu.au)
Chris Roberts (christopher.roberts@sydney.edu.au)
Tyler Clark (tyler.clark@sydney.edu.au)
Karyn Mossman (karyn.mossman@sydney.edu.au)

Version: 2 Date: 24 June 2014

Author's response to reviews:

Editor
BMC Medical Education

Title: The social validity of a national assessment centre for selection into general practice training.

We would like to submit the above manuscript for consideration for publication. Internationally, the process of recruiting the best candidates is central to the success of postgraduate training programs and the quality of the medical workforce. By considering social validity, the fairness and acceptable process to candidates can assessed. This paper, therefore, is relevant on an international level, and in line with current interests in selection processes for both postgraduate training and medical school selection.

Yours faithfully,
Annette Burgess